Impacts of socio-economic and climate changes on water, food, bioenergy, land use, and ecosystems

Tokuta Yokohata¹, Tsuguki Kinoshita², Gen Sakurai³, Shinichiro Fujimori⁴, Yadu Pokhrel⁵, Akihiko Ito¹, Yusuke Satoh¹, Etsushi Kato⁶, Masashi Okada¹, Kaoru Tachiiri⁷, Ken'ichi Matsumoto⁸, Seita Emori¹, and Kiyoshi Takahashi¹

¹National Institute for Environmental Studies
²Ibaraki University
³National Institute for Agro-Environmental Sciences
⁴Graduate School of Engineering, Kyoto University
⁵Michigan State University
⁶Institute of Applied Energy
⁷Japan Agency for Marine-Earth Science and Technology
⁸Toyo University

November 22, 2022

Abstract

Future socio-economic and climate changes can profoundly impact water resources, food production, bioenergy generation, and land use, leading to a broad range of societal problems. In this study, we performed future projections by using a land integrated model, MIROC-INTEG-LAND, that considers land surface physics, ecosystems, water management, crop growth, and land use, under various socio-economic scenarios (Shared Socio-economic Pathways, SSPs). Under the sustainability scenario (SSP1), demands for food and bioenergy are kept low, so that the increase in cropland areas for food and bioenergy are suppressed. On the contrary, in the middle of the road and regional rivalry scenarios (SSP2 and SSP3), cropland areas are projected to increase due to high demand for food and bioenergy. The expansion of cropland areas is projected to increase the water demand for irrigation and CO₂ emissions due to land use change. MIROC-INTEG-LAND simulations indicate that the impacts of the CO₂ fertilization effect and climate change on crop yields are comparable, with the latter being greater than the former under climate scenarios with high greenhouse gas concentrations. We also show that the CO₂ fertilization effects and climate change play important roles in changes in food cropland area, water demand for irrigation, and CO₂ emissions due to land use change. Our results underscore the importance of considering Earth-human system interactions when developing future socio-economic scenarios and studying climate change impacts.

1	Impacts of socio-economic and climate changes
2	on water, food, bioenergy, land use, and ecosystems
3	T. Yokohata ^{1*} , T. Kinoshita ² , G. Sakurai ³ , S. Fujimori ⁴ , Y. Pokhrel ⁵ , A. Ito ¹ , Y. Satoh ¹ , E.
4	Kato ⁶ , M. Okada ⁷ , K. Tachiiri ^{8,1} , K. Matsumoto ^{9,8} , S. Emori ¹ , and K. Takahashi ¹⁰
5	¹ Earth System Division, National Institute for Environmental Studies, Tsukuba, Japan
6	² College of Agriculture, Ibaraki University, Ami, Japan
7	³ Institute for Agro-Environmental Sciences, National Agriculture and Food Research
8	Organization, Tsukuba, Japan
9	⁴ Graduate School of Engineering, Kyoto University, Kyoto, Japan
10	⁵ Department of Civil and Environmental Engineering, Michigan State University, Michigan,
11	USA
12	⁶ Institute of Applied Energy, Tokyo, Japan
13	⁷ Climate Change Adaptation Center, National Institute for Environmental Studies, Tsukuba,
14	Japan
15	⁸ Research Institute for Global Change, Japan Agency for Marine-Earth Science and Technology,
16	Yokohama, Japan
17	⁹ Faculty of Economics, Toyo University, Tokyo, Japan
18	¹⁰ Social System Division, National Institute for Environmental Studies, Tsukuba, Japan
19	
20	Corresponding author: Takuta Vakahata (vakahata@nies go in)
20	Concesponding author. Tokuta Tokonata (<u>yokonata(a/mes.go.jp)</u>
21	
22	Key Points:
23	• Impacts of socio-economic and climate changes on water, food, bioenergy, land use and
24	ecosystems were analyzed using an integrated global land surface model.

- Cropland is projected to expand due to increased food demand and reduced crop yields
 caused by climate change, leading to an increased water demand and CO₂ emissions due
 to land use change.
- The CO₂ fertilization effects and climate change are projected to have considerable
 impacts on cropland area, water demand, and ecosystem production in future Earth human systems.
- 31

32 Abstract (fewer than 250 words, now 216 words)

Future socio-economic and climate changes can profoundly impact water resources, food 33 34 production, bioenergy generation, and land use, leading to a broad range of societal problems. In this study, we performed future projections by using a land integrated model, MIROC-INTEG-35 36 LAND, that considers land surface physics, ecosystems, water management, crop growth, and land use, under various socio-economic scenarios (Shared Socio-economic Pathways, SSPs). 37 Under the sustainability scenario (SSP1), demands for food and bioenergy are kept low, so that 38 the increase in cropland areas for food and bioenergy are suppressed. On the contrary, in the 39 40 middle of the road and regional rivalry scenarios (SSP2 and SSP3), cropland areas are projected to increase due to high demand for food and bioenergy. The expansion of cropland areas is 41 projected to increase the water demand for irrigation and CO₂ emissions due to land use change. 42 MIROC-INTEG-LAND simulations indicate that the impacts of the CO₂ fertilization effect and 43 climate change on crop yields are comparable, with the latter being greater than the former under 44 45 climate scenarios with high greenhouse gas concentrations. We also show that the CO_2 fertilization effects and climate change play important roles in changes in food cropland area, 46 water demand for irrigation, and CO₂ emissions due to land use change. Our results underscore 47 the importance of considering Earth-human system interactions when developing future socio-48 economic scenarios and studying climate change impacts. 49

50

51 Plain Language Summary

52 Future changes in society and climate will have a marked impact on water, food, energy, and 53 ecosystems, which can lead to a variety of problems in human society. In previous studies, these

issues have typically been investigated on a sector-by-sector basis because natural processes are 54 typically simplified in models that deal with human processes, and human processes are 55 simplified in models that deal with natural processes. We performed future projections using a 56 57 land integrated model, MIROC-INTEG-LAND, which combines models of global land surface physics, ecosystems, water management, crop growth, and land use. Our numerical simulations 58 showed that future changes in socio-economic conditions and climate are projected to impact 59 crop yields, resulting in substantial changes in future cropland area. The increase in cropland 60 61 area is projected to increase the water demand for irrigation and CO₂ emissions due to land use change. Our projections demonstrate the interconnections between the impacts of socioeconomic 62 63 and climate change on water, food, bioenergy, land use, and ecosystems, indicating the importance of making projections with models that describe the interactions between Earth and 64 65 human systems.

66

67 **1 Introduction**

68 Future changes in socio-economic factors and climate will have a major impact on human societies and natural ecosystems. It has been reported that the various climate impacts that occur 69 70 in a wide range of sectors interact with each other (Arent et al. 2014, Smajgl et al. 2016, Albrecht et al. 2018, Liu et al. 2018, Simpson et al. 2019, Yokohata et al. 2019, Naidoo et al. 2021). For 71 example, an increase in surface air temperatures and a decrease in rainfall due to climate change 72 can decrease available water resources (Cisneros et al. 2014, Schewe et al. 2014, Rodell et al. 73 74 2018, Ferguson et al. 2018, Pokhrel et al. 2021, Satoh et al. 2021). Changes in climate and water resources can reduce crop yields and affect food supply, trade, distribution, and prices, impacting 75 food availability (Schmidhuber and Tubiello 2007, Porter et al. 2014, Reisinger et al. 2014, 76 Niang et al. 2014, West 2014, Challinor et al. 2017, Mbow et al. 2019, Hasegawa et al. 2021). It 77 has also been demonstrated that the degradation of food security can cause health issues, such as 78 undernutrition and forced migration (Kniveton et al. 2012, Hsiang et al. 2013, Adger et al. 2014, 79 80 Smith et al. 2014, Mbow et al. 2019, Niles et al. 2021).

In studying issues related to the interactions between Earth and human systems, land use change caused by human activities is an important contributing factor to be considered (van Vuuren et al. 2012, Rounsevell et al. 2014, Lawrence et al. 2016, Popp et al. 2017, Jia et al.

2019, Yokohata et al. 2020). For example, meeting the increased food demand accompanying 84 future population growth will require an increase in crop yield and/or expansion in cropland area, 85 leading to shifts in land use (Foley et al. 2011, Weinzettel et al. 2013; Alexander et al. 2018). 86 87 Likewise, mitigating climate change by using bioenergy crops instead of fossil fuels requires vast areas of land for cultivating bioenergy crops (Smith et al. 2013, Humpenöder et al. 2015, Popp et 88 al. 2017, Alexander et al. 2018, Pete Smith et al. 2019, Muscat et al. 20209). Such expansion of 89 cropland area for food and bioenergy production can result in biodiversity loss (Immerzeel et al. 90 91 2014, Kehoe et al. 2017, Molotoks et al. 2018, Ohashi et al. 2019, Wu et al. 2019, Zabel et al. 2019), among a range of other impacts. In addition, conversion of forest to cropland can lead to 92 reduction in net carbon absorption by ecosystems (Brovkin et al. 2013, Lawrence et al. 2016, 93 Friedlingstein et al. 2020, Ito and Hajima 2020, Ito et al. 2020), leading to accelerated global 94 95 warming.

Studies based on integrated assessment and land use models have shown that there are 96 large uncertainties in future projections of land use changes, and the results of projections differ 97 markedly depending on the structure and assumptions of the models, and the interpretation of 98 future socio-economic storylines (Popp et al. 2017, Alexander et al. 2017). Factors such as 99 population, Gross Domestic Product (GDP), consumption trends, agricultural productivity, and 100 land use regulation are important determinants of future land use change (Stehfest et al. 2019). 101 Future climate change can also affect agricultural productivity, cropland land area, price, trade, 102 and consumption (Nelson et al. 2014, Wibe et al. 2015, Meijl et al. 2018). Further, the 103 104 implementation of climate mitigation measures such as bioenergy crop production to achieve the Paris goals can impact food consumption and increase the population at risk of hunger by raising 105 food prices (Fujimori et al. 2019, Hasegawa et al. 2015, 2018, 2020, 2021). 106

In the above studies investigating the impact of future socio-economic and climate changes on the agriculture sector, simulation results of crop yields are used mainly without considering CO₂ fertilization effects (Nelson et al. 2014, Wibe et al. 2015, Meijl et al. 2018, Hasegawa et al. 2018, 2020, 2021), primarily because there are large uncertainties associated with the latter (Wang et al. 2012, Boote et al. 2013). On the other hand, the latest crop modeling studies show that it is difficult to make realistic projections of crop yields without considering the CO₂ fertilization effects (Muller et al. 2021), and studies using results using both 'with' and 'without' CO₂ simulations interchangeably can potentially lead to misinterpretation of future
projections (Toreti et al. 2020).

In this study, we perform future projections by using MIROC-INTEG-LAND (MIROC 116 INTEgrated LAND surface model, Yokohata et al. 2020), which couples the latest global models 117 of land surface physics, water resources, crop growth, land use, and terrestrial ecosystems. In 118 previous studies, these impact assessments have been conducted separately on a sectoral basis 119 120 rather than by integrating natural and human components (Collins et al. 2015, Thornton et al. 2017, Muller-Hansen et al. 2017, Calvin and Bond-Lamberty 2018, Robinson et al. 2018, 121 Alexander et al. 2018, Monier et al. 2018, Tachiiri et al. 2021). MIROC-INTEG-LAND provides 122 a consistent, integrated modeling framework to examine these interconnections related to future 123 124 impacts. Taking advantage of this feature, we analyzed the impacts of climate change and CO_2 fertilization effects on crop growth, land use, water resources, and ecosystem production. In 125 addition, in order to investigate the sources of uncertainties in the future land use projections, we 126 compared the simulations of food and bioenergy cropland area simulated by MIROC-INTEG-127 LAND and those obtained using the integrated assessment model AIM/Hub (Fujimori et al. 128 2017), both of which have the same food and bioenergy demands. 129

In the reminder of this manuscript, section 2 provides an outline of MIROC-INTEG-130 LAND. Section 3 explains the setting of socio-economic and climate scenarios. Section 4 shows 131 the results of future projections estimated by MIROC-INTEG-LAND, focusing on the 132 interactions between climate, water resources, crops, land use, and ecosystems. Section 4 also 133 investigates the impact of climate change and CO₂ fertilization effects on crop yield, cropland 134 area, irrigation water, and ecosystems, and discusses the reasons for the differences between 135 136 MIROC-INTEG-LAND and AIM/Hub. Finally, Section 5 summarizes our findings and 137 important future work.

138

139 2 Model Description

Figure 1 shows the overall framework of the integrated land surface model MIROC-140 INTEG-LAND. The model was evaluated against observations made in our previous study 141 (Yokohata et al. 2020). Since MIROC-INTEG-LAND is based on the land surface component of 142 a global climate model MIROC (Model for Interdisciplinary Research on Climate version: 143 Watanabe et al., 2010), future climate scenarios are prescribed as the boundary conditions 144 (details in Section 3.2). In terms of socio-economic scenarios, GDP, the demands induced by 145 food (including feed) and bioenergy crop production, pasture area development, and timber 146 extraction (Terrestrial Land-use Model, Yokohata et al. 2020) are prescribed in the land use sub-147 model, TeLMO. The demands for domestic and industrial water abstraction are prescribed in the 148 water management sub-model, HiGWMAT (Pokhrel et al. 2015). In the crop growth sub-model, 149 PRYSBI2 (Sakurai et al. 2014), future technological developments factors that increase crop 150 yields are described as a function of GDP. For the data related to these socio-economic 151 scenarios, the results estimated by the integrated assessment model AIM/Hub (formerly 152 AIM/CGE) (Fujimori, Hasegawa et al. 2017) are used. 153



154

155 Figure 1. Structure of the global land integrated model MIROC INTEGrated LAND (MIROC-

156 INTEG-LAND, Yokohata et al. 2020). Colored boxes represent sub-models, and arrows indicate

variables that are exchanged. The names of the sub-models are also indicated in bold. The boxes
"Climate scenario" and "Socio-economic scenario" represent input data to the model. Modified

- 159 from Figure 1 in Yokohata et al. 2020.
- 160

Agricultural water demand was simulated in the water management sub-model 161 HiGWMAT (Pokhrel et al. 2015), withdrawn from rivers, reservoirs, and groundwater, and 162 added to the soil during the irrigation period. River flow regulation by dams is simulated using a 163 reservoir operation scheme in the HiGWMAT model. In irrigated grid cells, soil moisture and 164 165 evaporation amount in the land surface model MATSIRO (Minimal Advanced Treatments of Surface Interaction and Runoff, Takata et al. 2003, Nitta et al. 2014) are updated using an 166 irrigation scheme, and the calculated soil moisture and temperature are given to the crop growth 167 model PRYSBI2 (Sakurai et al. 2014). In PRYSBI2, five types of crops (spring and winter 168 wheat, rice, soybeans, and maize) are simulated over the global land area at a latitude/longitude 169 resolution of 1°. Key parameters that determine crop yield, such as the total number of heat units 170 until harvest (calculated based on degree days) and technological development factors, are 171 determined in each grid by using a statistical method (Sakurai et al. 2014) using historical global 172 yield data (Iizumi et al. 2014). The CO₂ fertilization effect in PRYSBI2 is formulated by using 173 the Farquhar model (Farquhar et al. 1980), and parameters of photosynthesis are calibrated 174 according to the observations (Sakurai et al. 2014, Yokohata et al. 2020). The decrease in crop 175 vields due to changes in temperature and water stress is calculated using PRYSBI2. 176

Crop yield simulated by PRYSBI2 is utilized by the land use model TeLMO (Yokohata 177 178 et al. 2020). In TeLMO, the areas of cropland for food and bioenergy production, pastureland, and natural and managed forest are calculated on a global 0.5° grid. In TeLMO, the food 179 180 cropland model has a cropland down-scaling module, which estimates an agricultural suitability index based on the crop yields obtained from PRYSBI2, 30-second resolution slope data, GDP of 181 182 17 regions around the world estimated by AIM/Hub, and food crop price (solved in TeLMO). In TeLMO, the area ratio of food cropland in the grid cell is more likely to be large if the 183 agricultural suitability index of a grid cell is large. The agricultural suitability index is 184 formulated based on the satellite observations of cropland area (Friedl et al. 2010). The food 185 cropland model in TeLMO also has an international trade module, which estimates prices in 17 186

regions around the world by solving a general equilibrium model in two sectors; the agriculturaland the non-agricultural sectors (Ejiri 2008).

An important feature of TeLMO is that the detailed spatial distribution of crop yield and 189 cropland distribution are considered when allocating the food cropland area. Specifically, crop 190 191 yield and cropland area in each grid cell are used to estimate the balance of the food supply and demand over 17 global regions using the general equilibrium model. Typically, integrated 192 193 assessment models use the average crop yield for each region to calculate the cropland area required for each region; consequently, it has been demonstrated that the spatial distribution of 194 195 crop yields cannot be fully reflected in the estimates of cropland area (Alexander et al. 2018). This issue is addressed in TeLMO by considering the spatial distribution of crop yield and 196 197 cropland area for each grid cell. In the TeLMO food cropland model, adjustment parameters are obtained and used so that the cropland area in the base year (2005) matches satellite observations 198 (Hurtt et al. 2020). 199

In the TeLMO bioenergy cropland model, the agricultural suitability index is formulated in a similar manner to the food cropland model. The price of bioenergy crops is adopted from the results of AIM/Hub. For the bioenergy crop yields, we used the yields of second-generation bioenergy crops (switchgrass, miscanthus) estimated by Kato and Yamagata (2014) using the SSP scenarios in Mori et al. (2018), and selected the higher yield value of the two crops for each grid.

TeLMO considers only bioenergy crop production as a climate mitigation measure. TeLMO calculates the decrease in forest area (deforestation) by considering timber demand, but does not consider afforestation or carbon capture and storage (CCS) as climate mitigation options. The development of forest management, afforestation, and CCS schemes are thus important areas of future work for model improvement.

The terrestrial ecosystem model VISIT (Vegetation Integrative SImulator for Trace gases, Ito & Inatomi 2012, Ito et al. 2018) uses the area ratio of the food and bioenergy crops, pasture, and forest (and transition matrix between them) calculated by TeLMO. In this way, the net CO₂ emissions associated with changes in land use can be estimated from the changes in the carbon balance (Ito 2019). In the VISIT model, plant photosynthesis is a function of ambient 216 CO₂ concentration, solar radiation, temperature, and soil moisture. Elevated CO₂ concentrations

217 promote the fertilization effect, but the strength of this effect varies with climatic conditions.

218

219 **3 Experimental Settings**

220 3.1 Socio-economic scenarios

In this study, combinations of multiple socio-economic and climate scenarios are used to project future changes in water resources, crops, land use, and ecosystems by MIROC-INTEG-LAND. We considered the following three socio-economic scenarios that are based on the SSP

framework (O'Neil et al. 2017, Riahi et al. 2017).

225 SSP1: Sustainability – Taking the green road

226 SSP2: Middle of the road

227 SSP3: Regional rivalry – A rocky road

Supplementary Table S1 summarizes the characteristics of each SSP as it applies to the estimates
 obtained using MIROC-INTEG-LAND. The characteristics of each socio-economic scenario are
 explained in Section 3.3.

231

232 3.2 Climate scenarios

In this study, daily bias-corrected climate model outputs from four models provided by 233 the Inter-Sectoral Impact Model Inter-comparison Project (ISIMIP, Hempel et al. 2013) are used 234 to drive the land surface physical process model (MATSIRO), crop growth model (PRYSBI2), 235 and terrestrial ecosystem model (VISIT). MATSIRO requires surface air temperature, 236 precipitation, wind speed, specific humidity, solar radiation, and surface pressure at 3-hourly 237 timesteps; the daily data are converted to 3-hour values based on the method proposed by Debele 238 et al. 2007 and Willet et al. 2007. The five climate models are GFDL-ES2M (Dunne et al., 239 2012), HadGEM2-ES (Jones et al., 2011), IPSL-CM5A-LR (Dufresne et al., 2012), Nor-ESM 240 241 (Bentsen et al., 2012), and MIROC-ESM-CHEM (Watanabe et al., 2011). For each model, four Representative Concentration Pathways (RCP2.6, 4.5, 6.0, 8.5, from 2006 to 2100) are used. 242

243

244

3.3 Combination of socio-economic and climate scenarios

Future projection experiments by MIROC-INTEG-LAND are conducted based on the 245 three socio-economic scenarios (SSP1-3) and four climate scenarios (RCP2.6, 4.5, 6.0, 8.5). As 246 explained above, MIROC-INTEG-LAND uses the results of the integrated assessment model 247 AIM/Hub as input, but in the calculations of AIM/Hub, the scenarios SSP1-RCP6.0 and SSP1-248 RCP8.5, SSP2-RCP8.5, SSP3-RCP8.5 are not available. This is because the greenhouse gas 249 (GHG) concentrations in the AIM/Hub baseline scenario are lower than RCP6.0 for SSP1 and 250 lower than RCP8.5 for SSP2 and SSP3. Therefore, in future projection simulations, the SSP1-251 baseline results estimated by AIM/Hub are used for SSP1-RCP6.0, those of the SSP2-baseline 252 for SSP2-RCP8.5, and those of the SSP3-baseline for SSP3-RCP8.5. It should be noted that there 253 is an unavoidable inconsistency between the input value estimated by AIM/Hub and other 254 estimates for SSP1-RCP6.0, SSP2-RCP8.5, and SSP3-RCP8.5. This inconsistency arises because 255 256 there is no bias-corrected climate scenario equivalent to the SSP1-, SSP2-, and SSP3-baselines in AIM/Hub. These inconsistencies are resolved when MIROC-INTEG-LAND is coupled with an 257 Earth System Models (ESMs) (Yokohata et al. in preparation) by calculating these climate 258 scenarios using the ESM component. 259

Figure 2 shows the time series of socio-economic scenarios, calculated by the integrated assessment model AIM/Hub. Population depends solely on SSPs (KC & Lutz 2017). In SSP1, which is a sustainability scenario, population growth is suppressed and the population decreases after 2050. Under SSP3, the population continues to grow, reaching around 13 billion by 2100. On the other hand, GDP increases significantly in SSP1 and SSP2, but not substantially in SSP3 (Dellink et al. 2017). GDP is hardly dependent on climate scenarios, partly because it does not consider the effects of climate change.

Future changes in food demand show a small increase in SSP1 and a large increase in SSP3, with SSP2 in between. Changes in food demand are not dependent on climate scenarios. On the other hand, future increases in the demand for bioenergy crops depend on both socioeconomic and climate scenarios. The demand for bioenergy crops is generally low for SSP1, as in the case of food demand. In SSP2, the demand for bioenergy crops under the RCP2.6, 4.5, 6.0 scenarios is large, larger than that in SSP1. The demand for bioenergy crops under RCP4.5, 6.0

in SSP3 is even greater than that in SSP2.

274



275

Figure 2. Time sequence of the sum of the global population (1st row), GDP (2nd row), food crop demand (3rd row), and bioenergy crop demand (4th row) under the SSP1 (left), SSP2 (middle),

and SSP3 (right) scenarios used as inputs to MIROC-INTEG-LAND. Population does not

depend solely on representative concentration pathways (RCPs), so it is shown in black. The line

colors show the results obtained for the RCP2.6 (blue), RCP4.5 (green), RCP6.0 (yellow), and

281 RCP8.5 (red) scenarios. See Section 3.3 for details on SSP-RCP combinations. Time sequences

are calculated using AIM/Hub.

283

284 **4 Results**

285

4.1 Changes in climate conditions

MIROC-INTEG-LAND uses atmospheric variables, such as surface air temperature and 286 287 precipitation, as inputs for the model, and calculates the land surface variables, such as soil moisture/temperature, and river flows, in the model. As shown in Figure S1, the variations in 288 surface air temperature rise and precipitation differ depending on the GHG concentration 289 scenarios. In the RCP8.5 scenario, where the surface air temperature rises the most, precipitation 290 increases, but average global soil moisture decreases. This is because evapotranspiration 291 increases globally as the surface air temperature rises. In addition, the future drought months in a 292 year are projected to increase as temperatures rise (Figure S2). Here, we defined the drought 293 months as the months when river flows are less than 20% of historical values. Figure S2 shows 294 that the number of drought months increases with the magnitude of climate change. In general, 295 296 future climate warming is projected to promote drying of the land surface in the climate model 297 ensemble used in this study.

298

299

4.2 Changes in crop yields

Figure 3 shows future changes in crop yields. The land-use model TeLMO uses the maximum yield of the five crops in each grid cell for the projection of cropland area (Yokohata et al. 2020). Figure 3 shows the time series for the maximum yields of the five crops in each grid cell averaged over the cropland area in the base year (2005). In the crop growth model PRYSBI2, future crop yields are generally determined by the CO₂ fertilization effects, climate change, and technological development. As shown in Figure 3, higher CO₂ concentrations in the climate scenario tend to have greater increases in crop yields; for example, the yields under RCP6.0 and RCP4.5 are higher than those under RCP2.6. On the other hand, under RCP8.5, the crop yield is projected to decrease in the latter half of the 21st century due to the large influence of climate change and the shortening of the growing period as the temperature rises (the growing period is estimated from total number of heat unit based on degree days; Yokohata et al. 2020).

Furthermore, in PRSYBI2, the increase in crop yield due to future technological developments is formulated as a function of GDP (Sakurai et al. 2014). Therefore, differences in the SSP scenarios also contribute to the future crop yields. Compared to SSP1 and SSP2, the effect of increasing yields through technological developments is smaller under the SSP3 scenario because GDP growth is slower.

Crop yields increase globally under RCP2.6, RCP4.5 and RCP6.0 (Figure S3). Under RCP8.5, the changes are mixed with a decrease in crop yields in the mid-latitudes (i.e., North America, Europe and inland Australia) and an increase in the low-latitudes (i.e., Africa and South America). The increase in crop yields is related to CO₂ fertilization effects and technological development, while the decrease is due to the effect of climate change (i.e., shortening of the growing period; Yokohata et al. 2020). A mechanistic description of the impacts of CO₂ fertilization effects and climate change on crop yields is provided in Section 5.



323



4.3 Changes in food and bioenergy cropland area

Figure 4a shows the future changes in food cropland area simulated by MIROC-INTEG-331 LAND. For comparison with the calculation results by MIROC-INTEG-LAND, Figure 4b also 332 shows the results of food cropland area by AIM/Hub (Fujimori et al. 2017). The MIROC-333 INTEG-LAND results under SSP1 show that the increase in food cropland area is small relative 334 to other SSPs because the demand for food crop is small under the SSP1 scenario (Figure 2). As 335 for their dependence on the climate scenario, the cropland area under RCP2.6 is relatively large 336 compared to that under RCP4.5 or RCP6.0, because the food crop yield is suppressed due to the 337 lower CO₂ fertilization effect. On the other hand, under RCP8.5, the food cropland area is 338 projected to increase in the latter half of the 21st century because the crop yields are projected to 339 decrease due to the impacts of climate change, as shown in Figure 3. With SSP3-RCP 8.5, the 340 food cropland area in 2100 is projected to expand to about twice that of the 20th century. Details 341 of the impacts of CO₂ fertilization effects and climate change on crop yields are discussed in 342 Section 5. 343



344

345

Figure 4. Time sequence of food cropland area calculated by MIROC-INTEG-LAND (top row).

347 The results of the SSP1 (left), SSP2 (middle), and SSP3 (right) scenarios. The cropland area is

calculated by AIM/Hub (bottom row). The unit is the ratio of cropland area to the global land

349 area. Line colors and styles are the same as those in Figure 3, but the results obtained using

- 350 AIM/Hub have only one ensemble member.
- 351

Figure S4 shows global maps of changes in the food cropland area estimated using 352 MIROC-INTEG-LAND. Under all SSPs, the food cropland area tends to increase in North and 353 354 South America. Changes in the food cropland area in Africa, India and China have different outcomes depending on the SSPs. Under SSP1, the food cropland area is projected to decrease in 355 Africa, India, and China. On the other hand, under SSP2 and SSP3, the food cropland area is 356 projected to increase in Africa, India and China, and the expansion is particularly large under 357 SSP3. In Africa, India and China, the changes in food cropland area differs markedly depending 358 on the population growth and food demand (Figure 2). 359

Figure 4b shows the time sequence of food cropland area estimated using AIM/Hub. As 360 for the dependency on socio-economic scenarios in AIM/Hub, the cropland area is projected to 361 decrease under SSP1, increase slightly under SSP2, and increase by up to 50% under SSP3 362 363 (Figure 4b). This dependency in the SSP scenarios should be interpreted as demand- and supplyside drivers. For the demand side, SSP1 is characterized by having a low population and low 364 365 consumption in livestock-related goods, which leads to relatively low agricultural production. In contrast, SSP3 has a larger population than the other SSPs and a relatively high livestock-related 366 367 demand; consequently, the cropland requirement is high. Considering the supply side, the crop yields under SSP1 are assumed to be higher than those under either SSP2 or SSP3, based on the 368 storylines of the SSPs (O'Neill et al. 2017), which can also be a factor contributing to the 369 decrease in cropland area in AIM/Hub. 370

As for the dependency of the climate scenario (RCPs) employed in AIM/Hub, the more mitigation measures are taken, the lower the expansion of food cropland area will be, as shown in Figure 4b. The climate policy factor in AIM/Hub is also associated with the supply-side stories. For example, land demand pressure would increase owing to an expansion in bioenergy demand and afforestation under low-carbon scenarios, such as RCP2.6. Consequently, land rent and crop yield could both increase due to such increases in land demand. Thus, both demandand supply-side changes would reduce cropland area under RCP2.6 (Figure 4b).

Figure 5a shows the future changes in bioenergy cropland area simulated by MIROC-INTEG-LAND. The results of the expansion in bioenergy cropland area differ markedly depending on the socio-economic scenario. As in the case of food cropland area, the demand for bioenergy crops is small under SSP1 compared to that under SSP2 and SSP3 (Figure 2), so the expansion of bioenergy cropland is small. As for the dependence on climate scenarios, bioenergy cropland areas are needed earlier and more extensively in scenarios where aggressive mitigation measures are implemented, such as those under RCP2.6 (Figure 5a).

385



386

Figure 5. The same as Figure 4, but graphs show the bioenergy cropland area.

388

Figure S5 shows the global distribution of changes in the bioenergy cropland area in MIROC-INTEG-LAND. The bioenergy cropland area is projected to increase in Africa, Australia, South America, and North America under all scenarios. Note that TeLMO assumes
that bioenergy crops are not grown in biodiversity hotspots (Yokohata et al. 2020). Changes in
bioenergy cropland area are greater in these areas under SSP2 and SSP3 compared to those under
SSP1, and under RCP2.6 and RCP4.5 compared to those under RCP8.5.

Figure 5b shows the results of the bioenergy cropland area estimated by AIM/Hub using 395 the same bioenergy crop demands as MIROC-INTEG-LAND. As in the case of food cropland 396 397 area, MIROC-INTEG-LAND requires a larger bioenergy cropland area compared to AIM/Hub, with the shape of the curves and the order of the scenarios being similar between the two models. 398 399 The results shown in Figure 5a and 5b are consistent with those of Kato and Yamagata (2014), who showed that the supply of bioenergy crops calculated using a process-based crop model is 400 401 larger than that calculated by integrated assessment models. This result is also affected by the fact that the biofuel crop model employed in Kato and Yamagata (2014) considers C4 herbaceous 402 plants which are relatively resistant to drying and high temperatures, and have small CO₂ 403 fertilization effects. 404

In AIM/Hub, the degree of bioenergy cropland expansion is dependent on pressure for decarbonization from the energy system side. The stringency of the climate mitigation targets would determine the scale of land required for bioenergy. Regarding the variations in SSP, again the energy system-side requirements would be a critical factor, as the SSP1 scenario is relatively dependent upon variable sources of renewable energy, such as solar and wind, whereas the SSP3 scenario is dependent on biomass. Taken together, these dependencies account for the observed differences in bioenergy cropland area (Figure 5b).

412 Figure 6a shows the sum of food and bioenergy cropland area estimated by MIROC-INTEG-LAND. As discussed above, differences in socio-economic scenarios play a major role 413 414 in the sum of food and bioenergy cropland area, with SSP1 keeping the increase in cropland area relatively small and stabilizing changes in cropland area in the middle of the 21st century. In 415 416 contrast, under the SSP2 and SSP3 scenarios, cropland area is projected to keep increasing throughout the 21st century. Under the SSP2 and SSP3 scenarios, the increase in demand for 417 bioenergy crops in RCP2.6, RCP4.5, and RCP6.0, and the decrease in food crop yield due to 418 climate change in RCP8.5, leading to an expansion in food cropland area. 419

420



Figure 6. The same as Figure 4, but graphs show the sum of food and bioenergy cropland areas.

Figure S6 shows the spatial distribution of changes in the sum of food and bioenergy 424 cropland areas in MIROC-INTEG-LAND. As mentioned earlier, under SSP1, the food cropland 425 area is projected to decrease in Africa and India, and the biofuel cropland area is projected to 426 increase slightly in Africa, Australia, South America, and North America. On the other hand, 427 under SSP2 and SSP3, the food cropland area is projected to increase in Africa, India and China, 428 and the bioenergy cropland area is projected to expand in Africa, Australia, South America, and 429 North America. In South and North America, under all socio-economic and climatic scenarios, 430 both food and bioenergy cropland area increases, and thus the increase in the sum of these 431 cropland areas becomes large. 432

Figure 6b shows that the sum of food and bioenergy cropland area estimated by AIM/Hub is smaller than the increase estimated by MIROC-INTEG-LAND. One of the reasons for the disparity between the two models is that they have different food and bioenergy crop yields. The 436 results of Figures 4 to 6 show that even if the future food and bioenergy demands are the same,

the results of the cropland area are projected to differ markedly depending on the crop yield and

the method used to estimate the cropland area. Details of the impacts of CO₂ fertilization effects

and climate change on food cropland area, as well as the differences in the future projections of

440 food cropland area between MIROC-INTEG-LAND and AIM/Hub are discussed in Section 5.

441 **4.**4

4.4 Changes in water demand and ecosystems

One consequence of the expansion of food cropland area is that there will be an increase 442 in water demand for irrigation. Figure 7a shows the irrigation water demand (the amount of 443 water abstracted from rivers and groundwater for irrigation) estimated by MIROC-INTEG-444 LAND. In general, increase (decrease) in food cropland area leads to increase (decrease) in water 445 demand, due to the increase (decrease) in irrigated cropland area (Yokohata et al. 2020). As 446 shown in Figure 4a, food cropland area expands in the second half of the 21st century due to the 447 lower crop yields, especially under the RCP8.5 scenario. As a result, water demand is projected 448 to increase, especially under the RCP8.5 scenario for SSP3, where the demand for food crops is 449 substantially high. On the other hand, SSP1 is projected to suppress the expansion of cropland 450 area, leading to reduced irrigation water demand. 451

Figure S7 shows the spatial distribution of changes in water demand for irrigation estimated using MIROC-INTEG-LAND. Water demand for irrigation is considered only in food cropland area, and the irrigation ratio (ratio of irrigated cropland area to total cropland area) of each grid cell is fixed at the value of the base year (2006). Under SSP1, the demand for irrigation water is projected to decrease in India and China due to the decrease in food cropland area. Under SSP2 and SSP3, the demand for irrigation water is projected to increase in India and China, where the food cropland area is increased, as shown in Figure 4a.

459



Figure 7. a) Time sequence of irrigation water demand. The unit is the global total annual irrigation water demand (unit is $km^3/year$); b) cumulative CO₂ emissions due to changes in land use (unit is GtCO₂, 10¹⁵ kg of CO₂). The cumulative values of the amount of CO₂ emitted in each year are shown. Line colors and styles are the same as those in Figure 3.

465

Figure 7b shows the cumulative CO₂ emissions due to changes in land use estimated by 466 the ecosystem sub-model VISIT. The cumulative CO₂ emissions due to the land use change were 467 estimated using the difference in land carbon content between the experiments with land use 468 change and those with fixed land use in the year 2005. Changes in land use are projected to 469 reduce net carbon sinks in terrestrial ecosystems. As shown in Figure 7b, SSP1 suppresses the 470 expansion of agricultural land, so cumulative CO₂ emissions can be kept relatively low 471 regardless of the climate scenario. The cumulative CO_2 emissions by 2100 due to land use 472 change under SSP1 are approximately 100 GtCO₂. On the other hand, under SSP2 and SSP3, the 473 cumulative CO₂ emissions due to changes in land use are projected to be larger compared to 474 SSP1. Not only under RCP2.6, where aggressive mitigation measures are adopted, but also under 475

476 RCP4.5 and RCP6.0, the cropland area is projected to expand due to increased demand for

- 477 bioenergy crops; as a result, cumulative CO₂ emissions increase. Overall, cropland area under
- 478 SSP3 is larger than that under SSP2, which would result in higher cumulative CO₂ emissions.
- 479 Cumulative CO₂ emissions by 2100 due to changes in land use are projected to be approximately
- 480 300 GtCO₂ under SSP2 (RCP2.6) and up to about 400 GtCO₂ under SSP3 (RCP4.5). According
- to the 6the Assessment report by Working Group 1 of Intergovernmental Panel for Climate
- 482 Change, the remaining carbon budget limiting global mean surface air temperature rises to 1.5
- and 2°C from pre-industrial levels with a 50% probability are 500 and 1350 GtCO₂, respectively
- (Canadell et al. 2021). The cumulative CO₂ emissions associated with land use change as shown
 in Figure 7b thus have a significant impact on the achievement of these climate stabilization
 targets.

Figure S8 shows the spatial distribution of cumulative CO₂ emissions from land use change simulated by MIROC-INTEG-LAND. The spatial distribution in Figure S8 is essentially similar to the changes in the sum of food and bioenergy cropland area shown in Figure S6. CO₂ emissions associated with land use change are higher under SSP2 and SSP3 than under SSP1, and are higher in Central Africa, South America, and North America.

492

493 **4.5 Impacts of CO₂ fertilization effects and climate change**

In this subsection, the impacts of CO₂ fertilization effects and climate change on food
crop yields, cropland area, irrigation water demand, and CO₂ emissions associated with land use
changes are discussed. We also discuss the differences in food cropland areas estimated using
MIROC-INTEG-LAND and AIM/Hub, as shown in Figures 4-6.

Figure 8a shows the crop yields of experiments where the climate condition is fixed (noCL+FE, using the 2006 value), those where the CO₂ fertilization effect is fixed (CL+noFE), and those where both of the climate condition and CO₂ fertilization effect are variable (CL+FE). In all experiments, the increase in crop yields due to technological developments (Sakurai et al. 2014) is considered. Note that the results of CL+FE are the same as those shown in Figure 3. As shown in Figure 8a, the crop yields in the noCL+FE experiments are higher than those

in the CL+FE experiments due to the absence of climate change impacts. Similarly, in the

- 505 CL+noFE experiments, the crop yields are smaller than those in the CL+FE experiment, because
- 506 there is no CO₂ fertilization effect. Figure 8 shows that the decrease in crop yields due to climate
- 507 change (climate impact: difference between CL+FE and noCL+FE, negative value) and the
- 508 increase in crop yields due to the CO₂ fertilization effects (CO₂ impact: difference between
- 509 CL+FE and CL+noFE, positive value) are comparable to those estimated by MIROC-INTEG-
- 510 LAND. The climate impacts are larger than the CO₂ impacts in scenarios with high CO₂
- 511 concentrations, such as the RCP8.5 scenario, leading to a decrease in crop yields in the second
- 512 half of the 21^{st} century.

513



Figure 8. Time sequence of a) crop yield (unit is t/ha) and b) food cropland area (unit is ratio to the global land area) in the experiments where the climate conditions are fixed at the year 2006 (noCL+FE: green), CO₂ fertilization effect is fixed (CL+noFE: cyan), and the impacts of climate change and CO₂ fertilization effect are considered (CL+FE: black).

519

514

Figure 8b shows the results of the food cropland area in the noCL+FE, CL+noFE, and 520 CL+FE experiments. Consistent with the results shown in Figure 8a, the cropland areas in the 521 noCL+FE experiment are smaller than those in the CL+FE experiments due to the higher crop 522 523 yields, and the cropland areas in the CL+noFE experiment are larger than those in the CL+FE experiment due to the lower crop yields. Interestingly, the absolute impacts of climate and CO₂ 524 on food crop yields are similar, but their impacts on food cropland areas are different. This is 525 because the food cropland area is generally inversely proportional to the food crop yields (see 526 527 details in Supporting Information Text).

Previous studies on the future impact of climate change on crop yields, land use changes, 528 and agricultural sectors have not considered the CO₂ fertilization effects (CL+noFE) (Nelson et 529 530 al. 2014, Wibe et al. 2015, Mijl et al. 2018, Hasegawa et al. 2015, 2018, 2021). As shown in Figure 8, in MIROC-INTEG-LAND, the CO₂ fertilization effect has comparable impact on crop 531 yields as that of climate change. Therefore, if the CO₂ fertilization effect is not taken into 532 consideration, the crop yield tends to be underestimated. Therefore, in a scenario with a large 533 CO₂ concentration increase such as under RCP8.5, not clarifying whether experiments were 534 performed 'with' or 'without' the CO₂ fertilization effect (CL+noFE and CL+FE) can result in 535 536 overestimating the future land use changes and their uncertainty ranges.

The food cropland areas simulated by AIM/Hub (Figure 4) do not consider the impacts of 537 climate change and CO₂ fertilization effects. Under the RCP2.6 and RCP4.5 scenarios, the 538 climate and CO₂ impacts on food crop yields are almost compensated for in MIROC-INTEG-539 LAND (Figure 8). Therefore, the reason why the cropland areas estimated by AIM/Hub are 540 smaller than those estimated by MIROC-INTEG-LAND are due to differences in technological 541 542 development or the effects of bioenergy demands which have the effect of increasing the food crop yields in AIM/Hub. On the other hand, under the RCP8.5 scenario, the impacts of climate 543 change decrease estimates of food crop yields in MIROC-INTEG-LAND, as the contribution of 544 545 climate change outweighs the CO_2 fertilization effects in the second half of the 21st century, as shown in Figure 8. One of the reasons why the cropland areas estimated by AIM/Hub are smaller 546 than those estimated by MIROC-INTEG-LAND under the RCP8.5 scenario is because climate 547 change is projected by the latter model to reduce food crop yields. 548

549

Finally, Figure 9 shows the water demand for irrigation and cumulative CO₂ emissions

associated with land use changes in the noCL+FE, CL+noFE, and CL+FE experiments. In the

absence of the CO₂ fertilization effect (CL+noFE), the food cropland area increases due to a

decrease in the crop yield, resulting in an increase in water demand for irrigation and CO₂

emissions associated with land use change. The results in Figure 9 show that the impacts of CO₂

fertilization effects and climate change play an important role in future changes in water demandand cumulative carbon emissions through the changes in cropland area.

556



Figure 9. Same as Figure 8, but for a) the water demand for irrigation (unit: $km^3/year$), and b) cumulative CO₂ emissions due to land use changes (unit: $GtCO_2 = 10^{15} kg$ of CO₂).

560

557

561

562 5 Conclusions

563 This study evaluated the impact of the combination of future socio-economic and climate 564 changes on water resources, crop growth, land use and ecosystems. Future food crop yields are 565 projected to increase, primarily due to CO₂ fertilization effects and technological developments; 566 however, they are projected to decrease when surface air temperatures rise substantially. Future

567 changes in food crop yields and demands are projected to affect changes in food cropland areas.

- 568 In addition, increasing demand for bioenergy crops for climate change mitigation can also lead to 569 the expansion of bioenergy cropland areas.
- Under the sustainability scenario (SSP1), increases in cropland area are kept small due to 570 low demand for food and bioenergy crops. On the other hand, under the middle of the road 571 572 (SSP2) and regional rivalry (SSP3) scenarios, the demand for food and bioenergy crops is higher than that under SSP1, so the increase in these cropland areas is large. In general, the 573 contributions of bioenergy cropland to total cropland area are projected to be substantial under 574 the RCP2.6, RCP4.5, and RCP6.0 scenarios, while those from food cropland are dominant under 575 576 the RCP8.5 scenario. The expansion of food and bioenergy cropland areas is projected to increase water demand for irrigation and CO₂ emissions associated with land use change. It is 577 important to create a green society, such as that envisaged under SSP1, by reducing the demand 578 for food and bioenergy and promoting technological innovations, in order to reduce CO₂ 579 emissions due to land use changes and to protect biodiversity. 580

Our analysis showed that the impacts of CO₂ fertilization effects and climate change on 581 crop yields are generally comparable, with the latter being larger than the former when the 582 increase in GHG concentrations are large such as under the RCP8.5 scenario. Our results also 583 showed that the CO₂ fertilization effects and climate change play a very important role in food 584 cropland areas, water demand for irrigation, and CO₂ emissions due to land use change. 585 Therefore, when the CO₂ fertilization effect is ignored, the food cropland areas are greatly 586 overestimated, especially under scenarios where the increase in CO₂ concentrations is large. To 587 588 improve the accuracy of projecting these impacts on the global system, it is important to increase current knowledge about the CO₂ fertilization effects as well as the impacts of climate change on 589 crop yields (Ainsworth and Long 2005, Erda et al. 2005, Sakurai et al. 2014, Degener 2015, 590 591 McGrath and Lobell 2013, Wang et al. 2020).

The socio-economic scenarios adopted in CMIP6 do not consider the impacts of climate change and CO₂ fertilization effects on crop yields when projecting future land use change (Riahi et al. 2017, O'Neil et al. 2017, van Vuuren et al. 2017, Fricko et al. 2017, <u>Fujimori</u> et al. 2017). The results obtained using MIROC-INTEG-LAND show that the expansion of cropland area is ⁵⁹⁶ larger than that estimated using AIM/Hub, which does not consider the impacts climate change

- ⁵⁹⁷ and the CO₂ fertilization effect. This difference between the two models is caused by the impacts
- of climate change, especially under scenarios with high GHG concentrations, such as RCP8.5.
- 599 The results of this study demonstrate the importance of considering the interactions between
- 600 Earth and human systems when constructing future socio-economic and climate scenarios.
- 601

602 Acknowledgments and Data

- 603 This research was supported by the "Integrated Research Program for Advancing Climate
- Models (TOUGOU Program)" sponsored by the Ministry of Education, Culture, Sports, Science,
- and Technology (MEXT), Japan (Grant Number JPMXD0717935715). This work was partially
- ⁶⁰⁶ supported by the Decarbonized and Sustainable Society Research Program at the National
- 607 Institute for Environmental Studies. Model simulations were performed on the SGI UV20 at the
- National Institute for Environmental Studies. The authors are much indebted to Keita
- 609 Matsumoto, Kuniyasu Hamada, and Taiga Matsushita for their support with model development,
- 610 numerical simulations, and compiling figures.
- 611

612 **Conflict of Interest**

- 613 The authors have no competing interests to declare.
- 614

615 Data Availability Statement

The MIROC-INTEG-LAND source code for this study is available for purposes of collaborative research with the model users under license from the copyright holders. For further information on how to obtain the code, please contact the corresponding author. The data from the model simulations and observations used in the analyses are available from the corresponding author.

621 **References**

622	Aggarwal, P., Vyas, S., Thornton, P., Campbell, B. M., & Kropff, M. (2019). Importance of
623	considering technology growth in impact assessments of climate change on agriculture.
624	Global Food Security, 23, 41-48. https://doi.org/https://doi.org/10.1016/j.gfs.2019.04.002
625	Ainsworth, E. A., & Long, S. P. (2005). What have we learned from 15 years of free-air CO2
626	enrichment (FACE)? A meta-analytic review of the responses of photosynthesis, canopy
627	properties and plant production to rising CO2. New Phytologist, 165(2), 351-372.
628	https://doi.org/doi:10.1111/j.1469-8137.2004.01224.x
629	Albrecht, T. R., Crootof, A., & Scott, C. A. (2018). The Water-Energy-Food Nexus: A
630	systematic review of methods for nexus assessment. Environmental Research Letters,
631	13(4), 043002. https://doi.org/10.1088/1748-9326/aaa9c6
632	Alexander, P., Prestele, R., Verburg, P. H., Arneth, A., Baranzelli, C., Batista e Silva, F., Brown,
633	C., Butler, A., Calvin, K., Dendoncker, N., Doelman, J. C., Dunford, R., Engström, K.,
634	Eitelberg, D., Fujimori, S., Harrison, P. A., Hasegawa, T., Havlik, P., Holzhauer, S.,
635	Humpenöder, F., Jacobs-Crisioni, C., Jain, A. K., Krisztin, T., Kyle, P., Lavalle, C.,
636	Lenton, T., Liu, J., Meiyappan, P., Popp, A., Powell, T., Sands, R. D., Schaldach, R.,
637	Stehfest, E., Steinbuks, J., Tabeau, A., van Meijl, H., Wise, M. A., & Rounsevell, M. D.
638	A. (2017). Assessing uncertainties in land cover projections. Global Change Biology,
639	23(2), 767-781. https://doi.org/10.1111/gcb.13447
640	Alexander, P., Rabin, S., Anthoni, P., Henry, R., Pugh, T. A. M., Rounsevell, M. D. A., &
641	Arneth, A. (2018). Adaptation of global land use and management intensity to changes in
642	climate and atmospheric carbon dioxide. Global Change Biology, 24(7), 2791-2809.
643	https://doi.org/10.1111/gcb.14110
644	Arent, D., Tol, R.S.J., Faust, E., Hella, J.P., Kumar, S., Strzepek, K.M., Tóth, F.L., et al. (2014).
645	Key Economic Sectors and Services. In: Climate Change 2014: Impacts, Adaptation, and
646	Vulnerability (Cambridge Univ. Press)
647	Boote, K. J., Jones, J. W., White, J. W., Asseng, S., & Lizaso, J. I. (2013). Putting mechanisms
648	into crop production models. Plant, Cell & Environment, 36(9), 1658-1672.
649	https://doi.org/https://doi.org/10.1111/pce.12119

650	Bentsen, M., Bethke, I., Debernard, J. B., Iversen, T., Kirkevåg, A., Seland, Ø., Drange, H.,
651	Roelandt, C., Seierstad, I. A., Hoose, C., & Kristjánsson, J. E. (2013). The Norwegian
652	Earth System Model, NorESM1-M – Part 1: Description and basic evaluation of the
653	physical climate. Geosci. Model Dev., 6(3), 687-720. https://doi.org/10.5194/gmd-6-687-
654	2013
655	Brovkin, V., Boysen, L., Arora, V. K., Boisier, J. P., Cadule, P., Chini, L., Claussen, M.,
656	Friedlingstein, P., Gayler, V., van den Hurk, B. J. J. M., Hurtt, G. C., Jones, C. D., Kato,
657	E., de Noblet-Ducoudré, N., Pacifico, F., Pongratz, J., & Weiss, M. (2013). Effect of
658	Anthropogenic Land-Use and Land-Cover Changes on Climate and Land Carbon Storage
659	in CMIP5 Projections for the Twenty-First Century. Journal of Climate, 26(18), 6859-
660	6881. https://doi.org/10.1175/jcli-d-12-00623.1
661	Calvin, K., & Bond-Lamberty, B. (2018). Integrated human-earth system modeling-state of the
662	science and future directions. Environmental Research Letters, 13(6), 063006.
663	https://doi.org/10.1088/1748-9326/aac642
664	Challinor, A. J., Adger, W. N., & Benton, T. G. (2017). Climate risks across borders and scales.
665	Nature Climate Change, 7(9), 621-623. https://doi.org/10.1038/nclimate3380
666	Cisneros, B. E., Oki, T., Arnell, N.W., Benito, G., Cogley, J. G., Döll, P., Jiang, T., et al. (2014).
667	Freshwater Resources. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability
668	(Cambridge Univ. Press)
669	Collins, W. D., Craig, A. P., Truesdale, J. E., Di Vittorio, A. V., Jones, A. D., Bond-Lamberty,
670	B., Calvin, K. V., Edmonds, J. A., Kim, S. H., Thomson, A. M., Patel, P., Zhou, Y., Mao,
671	J., Shi, X., Thornton, P. E., Chini, L. P., & Hurtt, G. C. (2015). The integrated Earth
672	system model version 1: formulation and functionality. Geosci. Model Dev., 8(7), 2203-
673	2219. https://doi.org/10.5194/gmd-8-2203-2015
674	Debele, B., Srinivasan, R., & Yves Parlange, J. (2007). Accuracy evaluation of weather data
675	generation and disaggregation methods at finer timescales. Advances in Water Resources,
676	30(5), 1286-1300. https://doi.org/https://doi.org/10.1016/j.advwatres.2006.11.009

677	Degener, J. (2015). Atmospheric CO2 fertilization effects on biomass yields of 10 crops in
678	northern Germany [Original Research]. Frontiers in Environmental Science, 3(48).
679	https://doi.org/10.3389/fenvs.2015.00048

Dellink, R., Chateau, J., Lanzi, E., & Magné, B. (2017). Long-term economic growth projections
in the Shared Socioeconomic Pathways. Global Environmental Change, 42, 200-214.
https://doi.org/https://doi.org/10.1016/j.gloenvcha.2015.06.004

Dufresne, J.-L., Foujols, M.-A., Denvil, S., Caubel, A., Marti, O., Aumont, O., Balkanski, Y.,

684 Bekki, S., Bellenger, H., Benshila, R., Bony, S., Bopp, L., Braconnot, P., Brockmann, P.,

685 Cadule, P., Cheruy, F., Codron, F., Cozic, A., Cugnet, D., de Noblet, N., Duvel, J.-P.,

686 Ethé, C., Fairhead, L., Fichefet, T., Flavoni, S., Friedlingstein, P., Grandpeix, J.-Y., Guez,

687 L., Guilyardi, E., Hauglustaine, D., Hourdin, F., Idelkadi, A., Ghattas, J., Joussaume, S.,

688 Kageyama, M., Krinner, G., Labetoulle, S., Lahellec, A., Lefebvre, M.-P., Lefevre, F.,

689 Levy, C., Li, Z. X., Lloyd, J., Lott, F., Madec, G., Mancip, M., Marchand, M., Masson,

690 S., Meurdesoif, Y., Mignot, J., Musat, I., Parouty, S., Polcher, J., Rio, C., Schulz, M.,

691 Swingedouw, D., Szopa, S., Talandier, C., Terray, P., Viovy, N., & Vuichard, N. (2013).

692 Climate change projections using the IPSL-CM5 Earth System Model: from CMIP3 to

693 CMIP5 [journal article]. Climate Dynamics, 40(9), 2123-2165.

694 https://doi.org/10.1007/s00382-012-1636-1

Dunne, J. P., John, J. G., Adcroft, A. J., Griffies, S. M., Hallberg, R. W., Shevliakova, E.,

696 Stouffer, R. J., Cooke, W., Dunne, K. A., Harrison, M. J., Krasting, J. P., Malyshev, S.

697 L., Milly, P. C. D., Phillipps, P. J., Sentman, L. T., Samuels, B. L., Spelman, M. J.,

698 Winton, M., Wittenberg, A. T., & Zadeh, N. (2012). GFDL's ESM2 Global Coupled

699 Climate–Carbon Earth System Models. Part I: Physical Formulation and Baseline

700 Simulation Characteristics. Journal of Climate, 25(19), 6646-6665.

701 https://doi.org/10.1175/jcli-d-11-00560.1

Ejiri, Y. (2008) A consideration of the Comparative Cost Model Using Three-Dimensional
 Diagrams, Forest Resource Management and Mathematical Modeling. FORMATH Vol.
 7, 7, 135-159.

705	Erda, L., Wei, X., Hui, J., Yinlong, X., Yue, L., Liping, B., & Liyong, X. (2005). Climate change
706	impacts on crop yield and quality with CO2 fertilization in China. Philos Trans R Soc
707	Lond B Biol Sci, 360(1463), 2149-2154. https://doi.org/10.1098/rstb.2005.1743
708	Farquhar, G. D., von Caemmerer, S. & Berry, J. A. (1980). A biochemical model of
709	photosynthetic CO2 assimilation in leaves of c3 species. Planta 149, 78–90.
710	Ferguson, C. R., Pan, M., & Oki, T. (2018). The Effect of Global Warming on Future Water
711	Availability: CMIP5 Synthesis [https://doi.org/10.1029/2018WR022792]. Water
712	Resources Research, 54(10), 7791-7819.
713	https://doi.org/https://doi.org/10.1029/2018WR022792
714	Foley, J. A., Ramankutty, N., Brauman, K. A., Cassidy, E. S., Gerber, J. S., Johnston, M.,
715	Mueller, N. D., O'Connell, C., Ray, D. K., West, P. C., Balzer, C., Bennett, E. M.,
716	Carpenter, S. R., Hill, J., Monfreda, C., Polasky, S., Rockström, J., Sheehan, J., Siebert,
717	S., Tilman, D., & Zaks, D. P. M. (2011). Solutions for a cultivated planet. Nature, 478,
718	337. https://doi.org/10.1038/nature10452
719	Fricko, O., Havlik, P., Rogelj, J., Klimont, Z., Gusti, M., Johnson, N., Kolp, P., Strubegger, M.,
720	Valin, H., Amann, M., Ermolieva, T., Forsell, N., Herrero, M., Heyes, C., Kindermann,
721	G., Krey, V., McCollum, D. L., Obersteiner, M., Pachauri, S., Rao, S., Schmid, E.,
722	Schoepp, W., & Riahi, K. (2017). The marker quantification of the Shared
723	Socioeconomic Pathway 2: A middle-of-the-road scenario for the 21st century. Global
724	Environmental Change, 42, 251-267.
725	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2016.06.004
726	Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Hauck, J., Olsen, A., Peters, G.
727	P., Peters, W., Pongratz, J., Sitch, S., Le Quéré, C., Canadell, J. G., Ciais, P., Jackson, R.
728	B., Alin, S., Aragão, L. E. O. C., Arneth, A., Arora, V., Bates, N. R., Becker, M., Benoit-
729	Cattin, A., Bittig, H. C., Bopp, L., Bultan, S., Chandra, N., Chevallier, F., Chini, L. P.,
730	Evans, W., Florentie, L., Forster, P. M., Gasser, T., Gehlen, M., Gilfillan, D., Gkritzalis,
731	T., Gregor, L., Gruber, N., Harris, I., Hartung, K., Haverd, V., Houghton, R. A., Ilyina,
732	T., Jain, A. K., Joetzjer, E., Kadono, K., Kato, E., Kitidis, V., Korsbakken, J. I.,
733	Landschützer, P., Lefèvre, N., Lenton, A., Lienert, S., Liu, Z., Lombardozzi, D., Marland,
734	G., Metzl, N., Munro, D. R., Nabel, J. E. M. S., Nakaoka, S. I., Niwa, Y., O'Brien, K.,

735	Ono, T., Palmer, P. I., Pierrot, D., Poulter, B., Resplandy, L., Robertson, E., Rödenbeck,
736	C., Schwinger, J., Séférian, R., Skjelvan, I., Smith, A. J. P., Sutton, A. J., Tanhua, T.,
737	Tans, P. P., Tian, H., Tilbrook, B., van der Werf, G., Vuichard, N., Walker, A. P.,
738	Wanninkhof, R., Watson, A. J., Willis, D., Wiltshire, A. J., Yuan, W., Yue, X., & Zaehle,
739	S. (2020). Global Carbon Budget 2020. Earth Syst. Sci. Data, 12(4), 3269-3340.
740	https://doi.org/10.5194/essd-12-3269-2020
741	Fujimori, S., Abe, M., Kinoshita, T., Hasegawa, T., Kawase, H., Kushida, K., Masui, T., Oka,
742	K., Shiogama, H., Takahashi, K., Tatebe, H., & Yoshikawa, M. (2017). Downscaling
743	Global Emissions and Its Implications Derived from Climate Model Experiments. PLoS
744	ONE, 12(1), e0169733. https://doi.org/10.1371/journal.pone.0169733
745	Fujimori, S., Hasegawa, T., Masui, T., Takahashi, K., Herran, D. S., Dai, H., Hijioka, Y., &
746	Kainuma, M. (2017). SSP3: AIM implementation of Shared Socioeconomic Pathways.
747	Global Environmental Change, 42, 268-283.
748	https://doi.org/http://dx.doi.org/10.1016/j.gloenvcha.2016.06.009
749	Fujimori, S., Hasegawa, T., Krey, V., Riahi, K., Bertram, C., Bodirsky, B. L., Bosetti, V., Callen,
750	J., Després, J., Doelman, J., Drouet, L., Emmerling, J., Frank, S., Fricko, O., Havlik, P.,
751	Humpenöder, F., Koopman, J. F. L., van Meijl, H., Ochi, Y., Popp, A., Schmitz, A.,
752	Takahashi, K., & van Vuuren, D. (2019). A multi-model assessment of food security
753	implications of climate change mitigation. Nature Sustainability, 2(5), 386-396.
754	https://doi.org/10.1038/s41893-019-0286-2
755	Hasegawa, T., Fujimori, S., Havlík, P., Valin, H., Bodirsky, B. L., Doelman, J. C., Fellmann, T.,
756	Kyle, P., Koopman, J. F. L., Lotze-Campen, H., Mason-D'Croz, D., Ochi, Y., Pérez
757	Domínguez, I., Stehfest, E., Sulser, T. B., Tabeau, A., Takahashi, K., Takakura, J. y., van
758	Meijl, H., van Zeist, WJ., Wiebe, K., & Witzke, P. (2018). Risk of increased food
759	insecurity under stringent global climate change mitigation policy. Nature Climate
760	Change, 8(8), 699-703. https://doi.org/10.1038/s41558-018-0230-x
761	Hasegawa, T., Sands, R. D., Brunelle, T., Cui, Y., Frank, S., Fujimori, S., & Popp, A. (2020).
762	Food security under high bioenergy demand toward long-term climate goals. Climatic
763	Change, 163(3), 1587-1601. https://doi.org/10.1007/s10584-020-02838-8

764	Hasegawa, T., Sakurai, G., Fujimori, S., Takahashi, K., Hijioka, Y., & Masui, T. (2021).
765	Extreme climate events increase risk of global food insecurity and adaptation needs.
766	Nature Food, 2(8), 587-595. https://doi.org/10.1038/s43016-021-00335-4
767	Hasegawa, T., Sakurai, G., Fujimori, S., Takahashi, K., Hijioka, Y., & Masui, T. (2021).
768	Extreme climate events increase risk of global food insecurity and adaptation needs.
769	Nature Food, 2(8), 587-595. https://doi.org/10.1038/s43016-021-00335-4
770	Hempel, S., Frieler, K., Warszawski, L., Schewe, J., & Piontek, F. (2013). A trend-preserving
771	bias correction – the ISI-MIP approach. Earth System Dynamics, 4(2), 219-236.
772	https://doi.org/10.5194/esd-4-219-2013
773	Hsiang, S. M., Burke, M., & Miguel, E. (2013). Quantifying the Influence of Climate on Human
774	Conflict. Science, 341(6151), 1235367. https://doi.org/10.1126/science.1235367
775	Humpenöder, F., Popp, A., Stevanovic, M., Müller, C., Bodirsky, B. L., Bonsch, M., Dietrich, J.
776	P., Lotze-Campen, H., Weindl, I., Biewald, A., & Rolinski, S. (2015). Land-Use and
777	Carbon Cycle Responses to Moderate Climate Change: Implications for Land-Based
778	Mitigation? Environmental Science & Technology, 49(11), 6731-6739.
779	https://doi.org/10.1021/es506201r
780	Hurtt, G. C., Chini, L., Sahajpal, R., Frolking, S., Bodirsky, B. L., Calvin, K., Doelman, J. C.,
781	Fisk, J., Fujimori, S., Klein Goldewijk, K., Hasegawa, T., Havlik, P., Heinimann, A.,
782	Humpenöder, F., Jungclaus, J., Kaplan, J. O., Kennedy, J., Krisztin, T., Lawrence, D.,
783	Lawrence, P., Ma, L., Mertz, O., Pongratz, J., Popp, A., Poulter, B., Riahi, K.,
784	Shevliakova, E., Stehfest, E., Thornton, P., Tubiello, F. N., van Vuuren, D. P., & Zhang,
785	X. (2020). Harmonization of global land use change and management for the period 850-
786	2100 (LUH2) for CMIP6. Geosci. Model Dev., 13(11), 5425-5464.
787	https://doi.org/10.5194/gmd-13-5425-2020
788	Iizumi, T., Okada, M., & Yokozawza, M. (2014). A meteorological forcing data set for global
789	crop modeling: Development, evaluation, and intercomparison. Journal of Geophysical
790	Research: Atmospheres, 119(2), 363-384. https://doi.org/10.1002/2013JD020130

791	Immerzeel D I Verweij P A van der Hilst F & Faaij A P C (2014) Biodiversity impacts
702	of bioenergy crop production: a state of the art review GCB Bioenergy 6(3) 183 200
792	https://doi.org/lations/10.1111/salk.12007
793	https://doi.org/https://doi.org/10.1111/gcbb.1206/
794	Ito, A. (2019), Disequilibrium of terrestrial ecosystem CO ₂ budget caused by disturbance-
795	induced emissions and non-CO2 carbon export flows: a global model assessment, Earth
796	System Dynamics, 10, 685-709, https://doi.org/10.5194/esd-10-685-2019.
797	Ito, A., & Hajima, T. (2020). Biogeophysical and biogeochemical impacts of land-use change
798	simulated by MIROC-ES2L. Progress in Earth and Planetary Science, 7(1), 54.
799	https://doi.org/10.1186/s40645-020-00372-w
800	Ito, A., Hajima, T., Lawrence, D. M., Brovkin, V., Delire, C., Guenet, B., Jones, C. D.,
801	Malyshev, S., Materia, S., McDermid, S. P., Peano, D., Pongratz, J., Robertson, E.,
802	Shevliakova, E., Vuichard, N., Wårlind, D., Wiltshire, A., & Ziehn, T. (2020). Soil
803	carbon sequestration simulated in CMIP6-LUMIP models: implications for climatic
804	mitigation. Environmental Research Letters, 15(12), 124061.
805	https://doi.org/10.1088/1748-9326/abc912
806	Ito, A., & Inatomi, M. (2012). Use of a process-based model for assessing the methane budgets
807	of global terrestrial ecosystems and evaluation of uncertainty. Biogeosciences, 9(2), 759-
808	773. https://doi.org/10.5194/bg-9-759-2012
809	Ito, A., Nishina, K., Ishijima, K., Hashimoto, S., & Inatomi, M. (2018). Emissions of nitrous
810	oxide (N2O) from soil surfaces and their historical changes in East Asia: a model-based
811	assessment. Progress in Earth and Planetary Science, 5(1), 55.
812	https://doi.org/10.1186/s40645-018-0215-4
813	Jia, G., E. Shevliakova, P. Artaxo, N. De Noblet-Ducoudré, R. Houghton, J. House, K. Kitajima,
814	C. Lennard, A. Popp, A. Sirin, R. Sukumar, L. Verchot, (2019) Land-climate
815	interactions. In: Climate Change and Land: an IPCC special report on climate change,
816	desertification, land degradation, sustainable land management, food security, and
817	greenhouse gas fluxes in terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia,
818	V. Masson-Delmotte, HO. Pörtner, D.C. Roberts, P. Zhai, R. Slade, S. Connors, R. van

819	Diemen, M. Ferrat, E. Haughey, S. Luz, S. Neogi, M. Pathak, J. Petzold, J. Portugal
820	Pereira, P. Vyas, E. Huntley, K. Kissick, M, Belkacemi, J. Malley, (eds.)].
821	Jones, C. D., Hughes, J. K., Bellouin, N., Hardiman, S. C., Jones, G. S., Knight, J., Liddicoat, S.,
822	O'Connor, F. M., Andres, R. J., Bell, C., Boo, K. O., Bozzo, A., Butchart, N., Cadule, P.,
823	Corbin, K. D., Doutriaux-Boucher, M., Friedlingstein, P., Gornall, J., Gray, L., Halloran,
824	P. R., Hurtt, G., Ingram, W. J., Lamarque, J. F., Law, R. M., Meinshausen, M., Osprey,
825	S., Palin, E. J., Parsons Chini, L., Raddatz, T., Sanderson, M. G., Sellar, A. A., Schurer,
826	A., Valdes, P., Wood, N., Woodward, S., Yoshioka, M., & Zerroukat, M. (2011). The
827	HadGEM2-ES implementation of CMIP5 centennial simulations. Geosci. Model Dev.,
828	4(3), 543-570. https://doi.org/10.5194/gmd-4-543-2011
829	Kato, E., & Yamagata, Y. (2014). BECCS capability of dedicated bioenergy crops under a future
830	land-use scenario targeting net negative carbon emissions. Earth's Future, 2(9), 421-439.
831	https://doi.org/10.1002/2014ef000249
832	Kc, S., & Lutz, W. (2017). The human core of the shared socioeconomic pathways: Population
833	scenarios by age, sex and level of education for all countries to 2100. Global
834	Environmental Change, 42, 181-192.
835	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2014.06.004
836	Kehoe, L., Romero-Muñoz, A., Polaina, E., Estes, L., Kreft, H., & Kuemmerle, T. (2017).
837	Biodiversity at risk under future cropland expansion and intensification. Nature Ecology
838	& Evolution, 1(8), 1129-1135. https://doi.org/10.1038/s41559-017-0234-3
839	Kniveton, D.R., Smith, C.D., & Black, R. (2012). Emerging migration flows in a changing
840	climate in dryland Africa. Nature Climate Change, 2, 444-447.
841	Lawrence, D. M., Hurtt, G. C., Arneth, A., Brovkin, V., Calvin, K. V., Jones, A. D., Jones, C. D.,
842	Lawrence, P. J., de Noblet-Ducoudré, N., Pongratz, J., Seneviratne, S. I., & Shevliakova,
843	E. (2016). The Land Use Model Intercomparison Project (LUMIP) contribution to
844	CMIP6: rationale and experimental design. Geosci. Model Dev., 9(9), 2973-2998.
845	https://doi.org/10.5194/gmd-9-2973-2016
846	Liu, J., Hull, V., Godfray, H. C. J., Tilman, D., Gleick, P., Hoff, H., Pahl-Wostl, C., Xu, Z.,
847	Chung, M. G., Sun, J., & Li, S. (2018). Nexus approaches to global sustainable

- development. Nature Sustainability, 1(9), 466-476. https://doi.org/10.1038/s41893-0180135-8
- Mbow, C., C. Rosenzweig, L.G. Barioni, T.G. Benton, M. Herrero, M. Krishnapillai, E. Liwenga, P. Pradh 850 an, M.G. Rivera-Ferre, T. Sapkota, F.N. Tubiello, Y. Xu (2019) Food Security. In: Climate 851 Change and Land: an IPCC special report on climate change, desertification, land 852 degradation, sustainable land management, food security, and greenhouse gas fluxes in 853 terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-854 O. Pörtner, D.C. Roberts, P. Zhai, R. Slade, S. Connors, R. van Diemen, M. Ferrat, E. 855 Haughey, S. Luz, S. Neogi, M. Pathak, J. Petzold, J. Portugal Pereira, P. Vyas, E. 856 Huntley, K. Kissick, M. Belkacemi, J. Malley, (eds.)]. 857 McGrath, J. M., & Lobell, D. B. (2013). Regional disparities in the CO 2 fertilization effect and 858 implications for crop yields. Environmental Research Letters, 8(1), 014054. 859 https://doi.org/10.1088/1748-9326/8/1/014054 860 861 Molotoks, A., Stehfest, E., Doelman, J., Albanito, F., Fitton, N., Dawson, T. P., & Smith, P. (2018). Global projections of future cropland expansion to 2050 and direct impacts on 862 biodiversity and carbon storage [https://doi.org/10.1111/gcb.14459]. Global Change 863 Biology, 24(12), 5895-5908. https://doi.org/https://doi.org/10.1111/gcb.14459 864 Monier, E., Paltsev, S., Sokolov, A., Chen, Y. H. H., Gao, X., Ejaz, Q., Couzo, E., Schlosser, C. 865 A., Dutkiewicz, S., Fant, C., Scott, J., Kicklighter, D., Morris, J., Jacoby, H., Prinn, R., & 866 Haigh, M. (2018). Toward a consistent modeling framework to assess multi-sectoral 867 climate impacts. Nat Commun, 9(1), 660. https://doi.org/10.1038/s41467-018-02984-9 868 Mori, S., Washida, T., Kurosawa, A., & Masui, T. (2018). Assessment of mitigation strategies as 869 tools for risk management under future uncertainties: a multi-model approach. 870 Sustainability Science, 13(2), 329-349. https://doi.org/10.1007/s11625-017-0521-6 871 Müller, C., Franke, J., Jägermeyr, J., Ruane, A. C., Elliott, J., Moyer, E., Heinke, J., Falloon, P. 872 D., Folberth, C., Francois, L., Hank, T., Izaurralde, R. C., Jacquemin, I., Liu, W., Olin, 873 S., Pugh, T. A. M., Williams, K., & Zabel, F. (2021). Exploring uncertainties in global 874 875 crop yield projections in a large ensemble of crop models and CMIP5 and CMIP6 climate

- scenarios. Environmental Research Letters, 16(3). https://doi.org/10.1088/17489326/abd8fc
- Müller-Hansen, F., Schlüter, M., Mäs, M., Donges, J. F., Kolb, J. J., Thonicke, K., & Heitzig, J.
 (2017). Towards representing human behavior and decision making in Earth system
 models an overview of techniques and approaches. Earth Syst. Dynam., 8(4), 977-1007.
 https://doi.org/10.5194/esd-8-977-2017
- Muscat, A., de Olde, E. M., de Boer, I. J. M., & Ripoll-Bosch, R. (2020). The battle for biomass:
 A systematic review of food-feed-fuel competition. Global Food Security, 25, 100330.
 https://doi.org/https://doi.org/10.1016/j.gfs.2019.100330
- Naidoo, D., Nhamo, L., Mpandeli, S., Sobratee, N., Senzanje, A., Liphadzi, S., Slotow, R.,
 Jacobson, M., Modi, A. T., & Mabhaudhi, T. (2021). Operationalising the water-energyfood nexus through the theory of change. Renewable and Sustainable Energy Reviews,
 149, 111416. https://doi.org/https://doi.org/10.1016/j.rser.2021.111416
- Nelson, G. C., Valin, H., Sands, R. D., Havlik, P., Ahammad, H., Deryng, D., Elliott, J.,
 Fujimori, S., Hasegawa, T., Heyhoe, E., Kyle, P., Von Lampe, M., Lotze-Campen, H.,
- ⁸³⁰ Fujihori, S., Hasegawa, T., Heynoe, E., Kyre, T., Von Lampe, W., Loize-Campen, H
- 891 Mason d'Croz, D., van Meijl, H., van der Mensbrugghe, D., Muller, C., Popp, A.,
- 892 Robertson, R., Robinson, S., Schmid, E., Schmitz, C., Tabeau, A., & Willenbockel, D.
- 893 (2014). Climate change effects on agriculture: economic responses to biophysical shocks.
- 894 Proc Natl Acad Sci U S A, 111(9), 3274-3279. https://doi.org/10.1073/pnas.1222465110
- Niang, I., Ruppel, O.C., Abdrabo, M.A., Essel, A., Lennard, C., Padgham, J., & Urquhart, P.
 (2014). Africa. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability
 (Cambridge Univ. Press)
- Niles, M. T., Emery, B. F., Wiltshire, S., Brown, M. E., Fisher, B., & Ricketts, T. H. (2021).
 Climate impacts associated with reduced diet diversity in children across nineteen
 countries. Environmental Research Letters, 16(1), 015010. https://doi.org/10.1088/17489326/abd0ab
- Nitta, T., Yoshimura, K., Takata, K., O'ishi, R., Sueyoshi, T., Kanae, S., Oki, T., Abe-Ouchi, A.,
 & Liston, G. E. (2014). Representing Variability in Subgrid Snow Cover and Snow

904	Depth in a Global Land Model: Offline Validation. Journal of Climate, 27(9), 3318-3330.
905	https://doi.org/10.1175/jcli-d-13-00310.1
906	O'Neill, B. C., Tebaldi, C., van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., Knutti,
907	R., Kriegler, E., Lamarque, J. F., Lowe, J., Meehl, G. A., Moss, R., Riahi, K., &
908	Sanderson, B. M. (2016). The Scenario Model Intercomparison Project (ScenarioMIP)
909	for CMIP6. Geosci. Model Dev., 9(9), 3461-3482. https://doi.org/10.5194/gmd-9-3461-
910	2016
911	O'Neill, B. C., Kriegler, E., Ebi, K. L., Kemp-Benedict, E., Riahi, K., Rothman, D. S., van
912	Ruijven, B. J., van Vuuren, D. P., Birkmann, J., Kok, K., Levy, M., & Solecki, W.
913	(2017). The roads ahead: Narratives for shared socioeconomic pathways describing world
914	futures in the 21st century. Global Environmental Change, 42, 169-180.
915	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2015.01.004
916	Ohashi, H., Hasegawa, T., Hirata, A., Fujimori, S., Takahashi, K., Tsuyama, I., Nakao, K.,
917	Kominami, Y., Tanaka, N., Hijioka, Y., & Matsui, T. (2019). Biodiversity can benefit
918	from climate stabilization despite adverse side effects of land-based mitigation. Nat
919	Commun, 10(1), 5240. https://doi.org/10.1038/s41467-019-13241-y
920	Pokhrel, Y., Felfelani, F., Satoh, Y., Boulange, J., Burek, P., Gädeke, A., Gerten, D., Gosling, S.
921	N., Grillakis, M., Gudmundsson, L., Hanasaki, N., Kim, H., Koutroulis, A., Liu, J.,
922	Papadimitriou, L., Schewe, J., Müller Schmied, H., Stacke, T., Telteu, CE., Thiery, W.,
923	Veldkamp, T., Zhao, F., & Wada, Y. (2021). Global terrestrial water storage and drought
924	severity under climate change. Nature Climate Change. https://doi.org/10.1038/s41558-
925	020-00972-w
926	Pokhrel, Y. N., Koirala, S., Yeh, P. J. F., Hanasaki, N., Longuevergne, L., Kanae, S., & Oki, T.
927	(2015). Incorporation of groundwater pumping in a global Land Surface Model with the
928	representation of human impacts. Water Resources Research, 51(1), 78-96.
929	https://doi.org/10.1002/2014WR015602
930	Popp, A., Calvin, K., Fujimori, S., Havlik, P., Humpenöder, F., Stehfest, E., Bodirsky, B. L.,
931	Dietrich, J. P., Doelmann, J. C., Gusti, M., Hasegawa, T., Kyle, P., Obersteiner, M.,
932	Tabeau, A., Takahashi, K., Valin, H., Waldhoff, S., Weindl, I., Wise, M., Kriegler, E.,

933	Lotze-Campen, H., Fricko, O., Riahi, K., & Vuuren, D. P. v. (2017). Land-use futures in
934	the shared socio-economic pathways. Global Environmental Change, 42, 331-345.
935	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2016.10.002
936	Popp, J., Lakner, Z., Harangi-Rákos, M., & Fári, M. (2014). The effect of bioenergy expansion:
937	Food, energy, and environment. Renewable and Sustainable Energy Reviews, 32, 559-
938	578. https://doi.org/https://doi.org/10.1016/j.rser.2014.01.056
939	Porter, J. R., Xie, L., Challinor, A.J., Cochrane, K., Howden, S.M., Iqbal, M.M., Lobell, D.B. et
940	al. (2014). Food Security and Food Production Systems. In: Climate Change 2014:
941	Impacts, Adaptation, and Vulnerability (Cambridge Univ. Press)
942	Riahi, K., van Vuuren, D. P., Kriegler, E., Edmonds, J., O'Neill, B. C., Fujimori, S., Bauer, N.,
943	Calvin, K., Dellink, R., Fricko, O., Lutz, W., Popp, A., Cuaresma, J. C., Kc, S.,
944	Leimbach, M., Jiang, L., Kram, T., Rao, S., Emmerling, J., Ebi, K., Hasegawa, T.,
945	Havlik, P., Humpenöder, F., Da Silva, L. A., Smith, S., Stehfest, E., Bosetti, V., Eom, J.,
946	Gernaat, D., Masui, T., Rogelj, J., Strefler, J., Drouet, L., Krey, V., Luderer, G.,
947	Harmsen, M., Takahashi, K., Baumstark, L., Doelman, J. C., Kainuma, M., Klimont, Z.,
948	Marangoni, G., Lotze-Campen, H., Obersteiner, M., Tabeau, A., & Tavoni, M. (2017).
949	The Shared Socioeconomic Pathways and their energy, land use, and greenhouse gas
950	emissions implications: An overview. Global Environmental Change, 42, 153-168.
951	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2016.05.009
952	Reisinger, A., Kitching, R.L., Chiew, F., Hughes, L., Newton, P.C.D., Schuster, S.S., Tait, A., et
953	al. (2014). Australasia. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability
954	(Cambridge Univ. Press)
955	Robinson, D. T., Di Vittorio, A., Alexander, P., Arneth, A., Barton, C. M., Brown, D. G.,
956	Kettner, A., Lemmen, C., O'Neill, B. C., Janssen, M., Pugh, T. A. M., Rabin, S. S.,
957	Rounsevell, M., Syvitski, J. P., Ullah, I., & Verburg, P. H. (2018). Modelling feedbacks
958	between human and natural processes in the land system. Earth Syst. Dynam., 9(2), 895-
959	914. https://doi.org/10.5194/esd-9-895-2018
960	Rodell, M., Famiglietti, J. S., Wiese, D. N., Reager, J. T., Beaudoing, H. K., Landerer, F. W., &
961	Lo, M. H. (2018). Emerging trends in global freshwater availability. Nature, 557(7707),
962	651-659. https://doi.org/10.1038/s41586-018-0123-1

963	Rounsevell, M. D. A., Arneth, A., Alexander, P., Brown, D. G., de Noblet-Ducoudré, N., Ellis,
964	E., Finnigan, J., Galvin, K., Grigg, N., Harman, I., Lennox, J., Magliocca, N., Parker, D.,
965	O'Neill, B. C., Verburg, P. H., & Young, O. (2014). Towards decision-based global land
966	use models for improved understanding of the Earth system. Earth Syst. Dynam., 5(1),
967	117-137. https://doi.org/10.5194/esd-5-117-2014
968	Sakurai, G., Iizumi, T., Nishimori, M., & Yokozawa, M. (2014). How much has the increase in
969	atmospheric CO2 directly affected past soybean production? Sci Rep, 4, 4978.
970	https://doi.org/10.1038/srep04978
971	Satoh, Y., Shiogama, H., Hanasaki, N., Pokhrel, Y. N., Boulange, J. E. S., Burek, P., Gosling, S.
972	N., Grillakis, M., Koutroulis, A. G., Müller Schmied, H., Thiery, W., & Yokohata, T.
973	(2021). A quantitative evaluation of the issue of drought definition: A source of
974	disagreement in future drought assessments. Environmental Research Letters.
975	https://doi.org/10.1088/1748-9326/ac2348
976	Schewe, J., Heinke, J., Gerten, D., Haddeland, I., Arnell, N. W., Clark, D. B., Dankers, R.,
977	Eisner, S., Fekete, B. M., Colón-González, F. J., Gosling, S. N., Kim, H., Liu, X.,
978	Masaki, Y., Portmann, F. T., Satoh, Y., Stacke, T., Tang, Q., Wada, Y., Wisser, D.,
979	Albrecht, T., Frieler, K., Piontek, F., Warszawski, L., & Kabat, P. (2014). Multimodel
980	assessment of water scarcity under climate change. Proceedings of the National Academy
981	of Sciences, 111(9), 3245-3250. https://doi.org/10.1073/pnas.1222460110
982	Schmidhuber, J., & Tubiello, F. N. (2007). Global food security under climate change.
983	Proceedings of the National Academy of Sciences, 104(50), 19703-19708.
984	https://doi.org/10.1073/pnas.0701976104
985	Simpson, G. B., & Jewitt, G. P. W. (2019). The Development of the Water-Energy-Food Nexus
986	as a Framework for Achieving Resource Security: A Review [Review]. Frontiers in
987	Environmental Science, 7(8). https://doi.org/10.3389/fenvs.2019.00008
988	Smajgl, A., Ward, J., & Pluschke, L. (2016). The water-food-energy Nexus - Realising a new
989	paradigm. Journal of Hydrology, 533, 533-540.
990	https://doi.org/https://doi.org/10.1016/j.jhydrol.2015.12.033

991	Smith, K.R., Woodward, A., Campbell-Lendrum, D., Chadee, D.D., Honda, Y., Liu, Q.,
992	Olwoch, J.M., et al. (2014). Human Health: Impacts, Adaption, and Co-Benefits. In:
993	Climate Change 2014: Impacts, Adaptation, and Vulnerability (Cambridge Univ. Press)
994	Smith, P., J. Nkem, K. Calvin, D. Campbell, F. Cherubini, G. Grassi, V. Korotkov, A.L. Hoang,
995	S. Lwasa, P. McElwee, E. Nkonya, N. Saigusa, JF. Soussana, M.A. Taboada (2019).
996	Interlinkages Between Desertification, Land Degradation, Food Security and Greenhouse
997	Gas Fluxes: Synergies, Trade-offs and Integrated Response Options. In: Climate Change
998	and Land: an IPCC special report on climate change, desertification, land degradation,
999	sustainable land management, food security, and greenhouse gas fluxes in terrestrial
1000	ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H O.
1001	Portner, D. C. Roberts, P. Zhai, R. Slade, S. Connors, R. van Diemen, M. Ferrat, E.
1002	Haughey, S. Luz, S. Neogi, M. Pathak, J. Petzold, J. Portugal Pereira, P. Vyas, E.
1003	Huntley, K. Kissick, M. Belkacemi, J. Malley, (eds.)].
1004	Smith, P., Haberl, H., Popp, A., Erb, Kh., Lauk, C., Harper, R., Tubiello, F. N., de Siqueira
1005	Pinto, A., Jafari, M., Sohi, S., Masera, O., Böttcher, H., Berndes, G., Bustamante, M.,
1006	Ahammad, H., Clark, H., Dong, H., Elsiddig, E. A., Mbow, C., Ravindranath, N. H.,
1007	Rice, C. W., Robledo Abad, C., Romanovskaya, A., Sperling, F., Herrero, M., House, J.
1008	I., & Rose, S. (2013). How much land-based greenhouse gas mitigation can be achieved
1009	without compromising food security and environmental goals? Global Change Biology,
1010	19(8), 2285-2302. https://doi.org/10.1111/gcb.12160
1011	Tachiiri, K., Su, X., & Matsumoto, K. i. (2021). Identifying key processes and sectors in the
1012	interaction between climate and socio-economic systems: a review toward integrating
1013	Earth-human systems. Progress in Earth and Planetary Science, 8(1), 24.
1014	https://doi.org/10.1186/s40645-021-00418-7
1015	Takata, K., Emori, S., & Watanabe, T. (2003). Development of the minimal advanced treatments
1016	of surface interaction and runoff. Global and Planetary Change, 38(1-2), 209-222.
1017	https://doi.org/10.1016/s0921-8181(03)00030-4
1018	Thornton, P. E., Calvin, K., Jones, A. D., Di Vittorio, A. V., Bond-Lamberty, B., Chini, L., Shi,
1019	X., Mao, J., Collins, W. D., Edmonds, J., Thomson, A., Truesdale, J., Craig, A.,
1020	Branstetter, M. L., & Hurtt, G. (2017). Biospheric feedback effects in a synchronously

1021	coupled model of human and Earth systems [Letter]. Nature Clim. Change, 7(7), 496-
1022	500. https://doi.org/10.1038/nclimate3310
1023	Toreti, A., Deryng, D., Tubiello, F. N., Müller, C., Kimball, B. A., Moser, G., Boote, K., Asseng,
1024	S., Pugh, T. A. M., Vanuytrecht, E., Pleijel, H., Webber, H., Durand, JL., Dentener, F.,
1025	Ceglar, A., Wang, X., Badeck, F., Lecerf, R., Wall, G. W., van den Berg, M., Hoegy, P.,
1026	Lopez-Lozano, R., Zampieri, M., Galmarini, S., O'Leary, G. J., Manderscheid, R.,
1027	Mencos Contreras, E., & Rosenzweig, C. (2020). Narrowing uncertainties in the effects
1028	of elevated CO2 on crops. Nature Food, 1(12), 775-782. https://doi.org/10.1038/s43016-
1029	020-00195-4
1030	van Meijl, H., Havlik, P., Lotze-Campen, H., Stehfest, E., Witzke, P., Domínguez, I. P.,
1031	Bodirsky, B. L., van Dijk, M., Doelman, J., Fellmann, T., Humpenöder, F., Koopman, J.
1032	F. L., Müller, C., Popp, A., Tabeau, A., Valin, H., & van Zeist, WJ. (2018). Comparing
1033	impacts of climate change and mitigation on global agriculture by 2050. Environmental
1034	Research Letters, 13(6). https://doi.org/10.1088/1748-9326/aabdc4
1035	van Vuuren, D. P., Batlle Bayer, L., Chuwah, C., Ganzeveld, L., Hazeleger, W., van den Hurk,
1036	B., van Noije, T., O'Neill, B., & Strengers, B. J. (2012). A comprehensive view on
1037	climate change: coupling of earth system and integrated assessment models.
1038	Environmental Research Letters, 7(2), 024012. https://doi.org/10.1088/1748-
1039	9326/7/2/024012
1040	van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., Hurtt, G.
1041	C., Kram, T., Krey, V., Lamarque, JF., Masui, T., Meinshausen, M., Nakicenovic, N.,
1042	Smith, S. J., & Rose, S. K. (2011). The representative concentration pathways: an
1043	overview. Climatic Change, 109(1-2), 5-31. https://doi.org/10.1007/s10584-011-0148-z
1044	van Vuuren, D. P., Stehfest, E., Gernaat, D. E. H. J., Doelman, J. C., van den Berg, M., Harmsen,
1045	M., de Boer, H. S., Bouwman, L. F., Daioglou, V., Edelenbosch, O. Y., Girod, B., Kram,
1046	T., Lassaletta, L., Lucas, P. L., van Meijl, H., Müller, C., van Ruijven, B. J., van der
1047	Sluis, S., & Tabeau, A. (2017). Energy, land-use and greenhouse gas emissions
1048	trajectories under a green growth paradigm. Global Environmental Change, 42, 237-250.
1049	https://doi.org/https://doi.org/10.1016/j.gloenvcha.2016.05.008

Wang, D., Heckathorn, S. A., Wang, X., & Philpott, S. M. (2012). A meta-analysis of plant 1050 physiological and growth responses to temperature and elevated CO2. Oecologia, 169(1), 1051 1-13. https://doi.org/10.1007/s00442-011-2172-0 1052 Wang, S., Zhang, Y., Ju, W., Chen, J. M., Ciais, P., Cescatti, A., Sardans, J., Janssens, I. A., Wu, 1053 1054 M., Berry, J. A., Campbell, E., Fernández-Martínez, M., Alkama, R., Sitch, S., Friedlingstein, P., Smith, W. K., Yuan, W., He, W., Lombardozzi, D., Kautz, M., Zhu, 1055 1056 D., Lienert, S., Kato, E., Poulter, B., Sanders, T. G. M., Krüger, I., Wang, R., Zeng, N., Tian, H., Vuichard, N., Jain, A. K., Wiltshire, A., Haverd, V., Goll, D. S., & Peñuelas, J. 1057 (2020). Recent global decline of CO₂ fertilization effects on vegetation 1058 photosynthesis. Science, 370(6522), 1295-1300. 1059 1060 https://doi.org/doi:10.1126/science.abb7772 Watanabe, M., Suzuki, T., O'ishi, R., Komuro, Y., Watanabe, S., Emori, S., Takemura, T., 1061 Chikira, M., Ogura, T., Sekiguchi, M., Takata, K., Yamazaki, D., Yokohata, T., Nozawa, 1062 T., Hasumi, H., Tatebe, H., & Kimoto, M. (2010). Improved Climate Simulation by 1063 MIROC5: Mean States, Variability, and Climate Sensitivity. Journal of Climate, 23(23), 1064 6312-6335. https://doi.org/10.1175/2010jcli3679.1 1065 1066 Watanabe, S., Hajima, T., Sudo, K., Nagashima, T., Takemura, T., Okajima, H., Nozawa, T., 1067 Kawase, H., Abe, M., Yokohata, T., Ise, T., Sato, H., Kato, E., Takata, K., Emori, S., & 1068 Kawamiya, M. (2011). MIROC-ESM 2010: model description and basic results of 1069 CMIP5-20c3m experiments. Geosci. Model Dev., 4(4), 845-872. https://doi.org/10.5194/gmd-4-845-2011 1070 1071 Weinzettel, J., Hertwich, E. G., Peters, G. P., Steen-Olsen, K., & Galli, A. (2013). Affluence drives the global displacement of land use. Global Environmental Change, 23(2), 433-1072 438. https://doi.org/https://doi.org/10.1016/j.gloenvcha.2012.12.010 1073 1074 West, P. C., Gerber, J. S., Engstrom, P. M., Mueller, N. D., Brauman, K. A., Carlson, K. M., 1075 Cassidy, E. S., Johnston, M., MacDonald, G. K., Ray, D. K., & Siebert, S. (2014). Leverage points for improving global food security and the environment. Science, 1076 345(6194), 325-328. https://doi.org/10.1126/science.1246067 1077

1078	Wiebe, K., Lotze-Campen, H., Sands, R., Tabeau, A., van der Mensbrugghe, D., Biewald, A.,
1079	Bodirsky, B., Islam, S., Kavallari, A., Mason-D'Croz, D., Müller, C., Popp, A.,
1080	Robertson, R., Robinson, S., van Meijl, H., & Willenbockel, D. (2015). Climate change
1081	impacts on agriculture in 2050 under a range of plausible socioeconomic and emissions
1082	scenarios. Environmental Research Letters, 10(8). https://doi.org/10.1088/1748-
1083	9326/10/8/085010
1084	Willett, K. M., Gillett, N. P., Jones, P. D., & Thorne, P. W. (2007). Attribution of observed
1085	surface humidity changes to human influence. Nature, 449(7163), 710-712.
1086	https://doi.org/10.1038/nature06207
1087	Wu, W., Hasegawa, T., Ohashi, H., Hanasaki, N., Liu, J., Matsui, T., Fujimori, S., Masui, T., &
1088	Takahashi, K. (2019). Global advanced bioenergy potential under environmental
1089	protection policies and societal transformation measures. GCB Bioenergy, 0(0).
1090	https://doi.org/10.1111/gcbb.12614
1091	Yokohata, T., Kinoshita, T., Sakurai, G., Pokhrel, Y., Ito, A., Okada, M., Satoh, Y., Kato, E.,
1092	Nitta, T., Fujimori, S., Felfelani, F., Masaki, Y., Iizumi, T., Nishimori, M., Hanasaki, N.,
1093	Takahashi, K., Yamagata, Y., & Emori, S. (2020). MIROC-INTEG-LAND version 1: a
1094	global biogeochemical land surface model with human water management, crop growth,
1095	and land-use change. Geosci. Model Dev., 13(10), 4713-4747.
1096	https://doi.org/10.5194/gmd-13-4713-2020
1097	Yokohata, T., Tanaka, K., Nishina, K., Takahashi, K., Emori, S., Kiguchi, M., Iseri, Y., Honda,
1098	Y., Okada, M., Masaki, Y., Yamamoto, A., Shigemitsu, M., Yoshimori, M., Sueyoshi, T.,
1099	Iwase, K., Hanasaki, N., Ito, A., Sakurai, G., Iizumi, T., Nishimori, M., Lim, W. H.,
1100	Miyazaki, C., Okamoto, A., Kanae, S., & Oki, T. (2019). Visualizing the
1101	Interconnections Among Climate Risks. Earth's Future, 7(2), 85-100.
1102	https://doi.org/10.1029/2018ef000945
1103	Zabel, F., Delzeit, R., Schneider, J. M., Seppelt, R., Mauser, W., & Václavík, T. (2019). Global
1104	impacts of future cropland expansion and intensification on agricultural markets and
1105	biodiversity. Nat Commun, 10(1), 2844. https://doi.org/10.1038/s41467-019-10775-z



Earth's Future

Supporting Information for

Impacts of socio-economic and climate changes

on water, food, bioenergy, land use, and ecosystems

T. Yokohata^{1*}, T. Kinoshita², G. Sakurai³, S. Fujimori⁴, A. Ito¹, Y. Satoh¹, Y. Pokhrel⁵, E. Kato⁶, M. Okada⁷, K. Tachiiri^{8,1}, K. Matsumoto^{9,8}, S. Emori¹, and K. Takahashi¹⁰

1 Earth System Division, National Institute for Environmental Studies, Tsukuba, Japan

2 College of Agriculture, Ibaraki University, Ami, Japan

3 Institute for Agro-Environmental Sciences, National Agriculture and Food Research Organization, Tsukuba, Japan

4 Graduate School of Engineering, Kyoto University, Kyoto, Japan

5 Department of Civil and Environmental Engineering, Michigan State University, Michigan, USA

6 Institute of Applied Energy, Tokyo, Japan

7 Climate Change Adaptation Center, National Institute for Environmental Studies, Tsukuba, Japan

8 Research Institute for Global Change, Japan Agency for Marine-Earth Science and Technology, Yokohama, Japan

9 Faculty of Economics, Toyo University, Tokyo, Japan

10 Social System Division, National Institute for Environmental Studies, Tsukuba, Japan

Contents of this file

Text S1 Captions for Tables S1 Figure S1 to S8

Additional Supporting Information (Files uploaded separately)

Tables S1

Text S1. The impacts of climate and CO2 on food crop yields and cropland area

As shown in Figure 8, the absolute impacts of climate and CO2 on food crop yields are similar, but their impacts on food cropland areas are different. This is because the food cropland area is generally inversely proportional to the food crop yields. Let Y_0 be the crop yield of the CL+FE experiment, $Y_0 + Y_{CL}$ be the crop yield of the noCL+FE experiment, and Y_0-Y_{FE} be the crop yield of the CL+noFE experiment. Here, Y_{CL} and Y_{FE} represent the contributions of the climate change and the CO₂ fertilization effect on the crop yield, respectively (both positive values). Assuming that the food cropland areas of the CL+FE, noCL+FE, and CL+noFE experiments are A₀, A_{CL}, and A_{FE}, respectively, then the ratio of each cropland area is as follows.

$$A_{CL}/A_0 \sim Y_0/(Y_0+Y_{CL}) = (1+Y_{CL}/Y_0)^{-1}$$

 $A_{FE}/A_0 \sim Y_0/(Y_0-Y_{FE}) = (1-Y_{FE}/Y_0)^{-1}$

If the contributions of climate change and the CO_2 fertilization effects are very small (Y_{CL}/Y_0 , Y_{FE}/Y_0 approach to zero),

$$A_{CL}/A_0 \simeq 1$$
,
 $A_{FE}/A_0 \simeq 1$.

On the other hand, if the contributions of climate change and the CO_2 fertilization effect are large enough, that is, if Y_{CL}/Y_0 and Y_{FE}/Y_0 approach to 1, then

$$A_{CL}/A_0 \rightarrow 1/2,$$

 $A_{FF}/A_0 \rightarrow \infty.$

Therefore, the contribution of CO₂ fertilization effects on food cropland area (A_{FE}) is relatively large compared to that of climate change (A_{CL}). For example, if the food crop yields in 2100 under the RCP8.5 scenario are Y₀ = 5.5, Y_{CL} = 4.0, Y_{FE} = 3.0 [t/ha], then the changes in food cropland area can be estimated as A_{CL}/A₀ ~ 0.57 and A_{FE}/A₀ ~ 2.2 by using the above equations. This estimation corresponds to a decrease in A_{CL} by approximately 43%, A_{FE} increases by approximately 120%, which is generally consistent with the results shown in Figure 9. In other words, even if the climate and CO₂ impacts on crop yields (Y_{CL} and Y_{FE}) are comparable, their effects on changes in cropland areas (A_{CL} - A_0 and A_{FE} - A_0) are relatively larger in the latter case compared to the former case.

Table S1. The characteristics of the SSP scenarios adopted from O'Neil et al. 2017.







Figure S2. Time sequence of the number of drought months calculated by MIROC-INTEG-LAND. The average results over the global land area are shown. Drought months are defined as the number of months with river flows below the 20 percentiles for each month in historical experiments (1950-2005). If the number of drought months is 3, it means that there are 3 months in the year with river flows below the 20



Figure S3. Global map showing anomalies in crop yields (unit is tons/ha). The anomalies are calculated by the difference between the averages of 2081-2100 and 2006-2025. Crop yield is calculated in the same way as in Figure 3.



Figure S4. Same as Figure S3, but for anomalies in food cropland area. The unit is the ratio of the cropland area in each grid / total area of the grid.



Figure S5. Same as Figure S3, but for anomalies in bioenergy cropland area. The unit is the same as that in Figure S4.



Figure S6. Same as Figure S3, but for anomalies in bioenergy cropland area. The unit is the same as that in Figure S4.



Figure S7. Same as Figure S3, but for anomalies in water demand for irrigation. The unit is kg/sec in each grid (1° longitude and latitude).



Figure S8. Time Cumulative CO_2 emissions due to land use change. The cumulative emission from 2006 to 2100 is shown. The unit is kgCO₂/m².